Author's response to reviews

Title: HIV-1 Tat protein alter the tight junction integrity and function of retinal pigment epithelium: an in vitro study

Authors:

Ling Bai (larkling@163.com)
Zhenping Zhang (zhenpin.z@hotmail.com)
Hui Zhang (hui.zhang@jefferson.edu)
Xiumei Li (Xium.lee@gmail.com)
Qiurong Yu (QiuR.yu@gmail.com)
Haotian Lin (haot.lin@hotmail.com)
Wenhui Yang (wenhui.yg@gmail.com)

Version: 5 Date: 4 June 2008

Author's response to reviews: see over
Dear editor,

Thank you for your message concerning our manuscript entitled "HIV-1 Tat Protein Alter the Tight Junction Integrity and Function of Retinal Pigment Epithelium: an in vitro Study" (BMC Infectious Diseases Ms. No.: 1375774087181044).

We are glad to know that our manuscript will be published in BMC Infectious Diseases. I specify changes made in the manuscript below, replying point-by-point to the editor’s suggestion.

*Formatting changes requested:*

1) Title - Please use sentence case in the title (capitalise only the first word, and proper nouns), both in the manuscript file and the submission system (i.e. remove all unnecessary capitals).

   **Answer:** I have changed the title.

2) Each email address should be on a new line and should be preceded by the author's initials, not the full name.

   **Answer:** I have changed the author's full name into initials.

3) List of abbreviations used -> Abbreviations

   **Answer:** I have changed it as you suggested

Thank you for your hard works again. Publication in BMC Infectious Diseases, it’s our great honor and it would be the great help for Dr Ling Bai, the first author, to apply her Ph.D. degree. We really appreciate that this manuscript could be available online before 6 Jun, because if this manuscript published later than that date, her getting Ph.D. degree will be delayed to Jan 2009. Please do not hesitate to contact us if there is something we can do.

Sorry for the inconvenient to your job.

Sincerely,

Zhenping Zhang